Author's response to reviews

Title: Double auricles of the right atrium: a unique anatomic deformity

Authors:

Georgios I. Tagarakis (gtagarakis@gmail.com)
Dimos Karangelis (dimoskaragel@yahoo.gr)
Marios E. Daskalopoulos (daskalopoulos@gmail.com)
Dimitrios Papadopoulos (dcpapadopoulosmd@gmail.com)
Theocharis Koufakis (thkoyfak@med.uth.gr)
Ioannis Karantzis (karantzis@hotmail.com)
Stefania S Lampoura (slampoura@gmail.com)
Andony J Baddour (badour@otenet.gr)
Nikolaos B Tsilimingas (tsilimingas@med.uth.gr)

Version: 2 Date: 26 January 2011

Author's response to reviews:

Dear editor,

We wish to thank you for giving us the opportunity to revise our manuscript. In the revised one we tried to conform with all the reviewers requests so that the manuscript meets the criteria for publication by your esteemed journal. All changes are marked in red. More specifically we have structured the abstract according to your directions and asked a native speaker to check the manuscript. We have also included an Acknowledgements section as instructed. No sources of funding were necessary for this report. Moreover, we have responded with a point-by point answers section to the reviewers’ comments.

Reviewer R. Hetzer: We agree with the the reviewer that the presented deformity is of minor clinical significance though scientifically interesting. We have corrected the reference as asked.

Reviewer Rainer G.H. Moosdorf : We have include arrows in the image as per reviewer's request and we have added an echo image.

Reviewer Filipos Triposkiadis : We added an echocardiography image and commented respectively.

Reviewer Farhood FS Saremi : We have commented on the exclusion of the diagnosis of right juxtaposition, added an echocardiography image and enriched the basic picture with arrows.

Kind regards,
Dimos Karangelis MD, PhD